Author’s response to reviews

Title: Barriers to and Competency with the Use of Metered Dose Inhaler and Its Impact on Disease Control among Adult Asthmatic Patients in Ethiopia

Authors:

Tadesse Melaku Abegaz (abegaztadesse981@gmail.com)

Efrata Shegena (ashuroefrata@gmail.com)

Natnael Gessie (melakutadesse981@gmail.com)

Eyob Gebreyohanns (justeyob@gmail.com)

Mohammed Seid (hassenm100@gmail.com)

Version: 2 Date: 06 Feb 2020

Author’s response to reviews:

Authors’ reply to the editor

1. If consent was obtained verbally please include reasoning for this and state if an IRB approved this method of obtaining consent.

Dear editor

Thank you for your comment

The consent was obtained both verbally and in written form

2. We note that only summarized data is presented in this manuscript, so we request that you amend this statement. Submission of a manuscript to a BMC Series journal implies that materials described in the manuscript, including all relevant raw data, will be freely available to any scientist wishing to use them for non-commercial purposes, without breaching participant confidentiality.
Please detail where the raw data supporting your findings can be found (including information found in any supplementary files). If the raw data is publicly available or can be requested from a particular author, please state that this is the case, and where the data can be found or requested from. Alternatively, if you do not wish to or cannot share your data, please state (in the ‘Availability of data and Materials’ section) that data will not be shared, and state the reason.

3. Dear editor

Thank you again for your comment,

We included the following statement in the availability of data.

“The data that support the findings of this study are available on request from the corresponding author, [TMA].”

3. Please remove the response to reviewers.

We remove response to reviewers.